Author's response to reviews

Title: More rumination and less effective emotion regulation in previously depressed individuals with preserved executive functions

Authors:

  Martin Aker (martin.aker@psykologi.uio.no)
  Catherine Harmer (catherine.harmer@psych.ox.ac.uk)
  Nils Inge Landrø (n.i.landro@psykologi.uio.no)

Version: 2
Date: 18 August 2014

Author's response to reviews:

Following a request from the editorial office, the following changes were made:

1. Added page numbers
2. Added line numbers
3. The ethics statement, including written consent information, was moved from the Procedure subsection at the end of Methods to a separate subsection first in Methods.